Author’s response to reviews

Title: Taxonomy of approaches to developing interventions to improve health: a systematic methods overview

Authors:

Alicia O'Cathain (a.ocathain@sheffield.ac.uk)

Liz Croot (l.croot@sheffield.ac.uk)

Katie Sworn (k.sworn@sheffield.ac.uk)

Edward Duncan (edward.duncan@stir.ac.uk)

Nikki Rousseau (nikki.rousseau@stir.ac.uk)

Katrina Turner (katrina.turner@bristol.ac.uk)

Lucy Yardley (lucy.yardley@bristol.ac.uk)

Pat Hoddinott (p.m.hoddinott@stir.ac.uk)

Version: 2 Date: 17 Feb 2019

Author’s response to reviews:

Dear Dr Jinhui Ma

Many thanks for your further comments.

We are pleased that you consider the paper to be scientifically sound and potentially acceptable for publication.

We have specifically addressed the three grammatical issues you identified. For page 5 line 112-114 we did not use your suggestion because this changed the meaning of the sentence. We did however make edits to ensure the sentence conveyed the intended meaning.

We asked an ex-editor to read the manuscript and sub-edit it. The further edits shown by track changes have been made by him. He has spent many years publishing academic papers as well as editing the work of others.

If you do accept the paper, we are holding a conference of 100 people interested in intervention development on March 25th where we would like to publicise this paper.
Best wishes

Alicia O’Cathain on behalf of all authors